Author's response to reviews

Title: Multiple metastasis of soft tissue visualized by 99mTc MDP scintigraphy: a case report.

Authors:

Mauro Liberatore (maurlib@tin.it)
Valentina Megna (megna_valentina@virgilio.it)
Gregorio Patrizi (g_patrizi@yahoo.com)
Domenico Giannotti (dome.46@hotmail.it)
Stefania Rebonato (stefyrebon@libero.it)
Flavio Barchetti (flavio.barchetti@live.it)
Pietro Gallo (pietro.gallo@uniroma1.it)
Giandomenico Miscusi (giandomenico.miscusi@uniroma1.it)

Version: 2
Date: 2 May 2014

Author's response to reviews: see over
Dear Editor-in-Chief,

I am sending you our manuscript entitled “Multiple metastasis of soft tissue visualized by 99mTc MDP scintigraphy: a case report.” by Mauro Liberatore, Valentina Megna, Gregorio Patrizi, Domenico Giannotti, Stefania Rebonato, Flavio Barchetti, Pietro Gallo and Giandomenico Miscusi. We would like to have the manuscript considered for publication in the Journal of Medical Case Reports.

We confirm that this manuscript has not been published elsewhere and is not under consideration by another journal.

All authors have approved the manuscript and agree with its submission to Journal of Medical Case Reports.

As far as we know, our case is very rare and might interest different specialties of doctors: from surgeons to oncologists and pathologists, from radiologists to radiation oncologists.

It describes different imaging techniques of the ossification in soft tissues from colorectal cancer and the managing of such case, which is an unusual presentation (soft tissue metastases) of a very common disease (colorectal cancer).

Written informed consent was obtained from the patient’s next of kin for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal.

The authors have no potential competing interests.

I hope it will meet your interest and your editorial needs.

Best regards,

Gregorio Patrizi